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Notes for the March 5, 2014, CERES  

Processing Oversight Board Meeting  

 

Meeting began at 1:00 p.m. in B1268C R2316. 

Meeting Notes  

There were no comments or questions on the notes from the last meeting.   

Announcements 

None. 

    

SCCR Discussion (SCCRs that have been submitted or updated since the last meeting) 

Synoptic SARB 1002 – Using CLASS Mesh 8 snow/ice maps instead of NOAA for April 2013 

and forward.  Delivering an update soon to look for MARCH files in the DPO instead of the 

subsystem directory. 

 

There were no issues identified, so this SCCR will be approved. 

 

CATALYST Status 

The schedule was updated last week, but Tonya still needs to review it.  Two days will be added 

to account for the recent snow days.  Expected go-live date is in early April. 

   

Writing up the RFAs.  Chris and Tonya are each addressing one and Jonathan is addressing the 

other two. 

   

The data required for the 8 PGEs ready to run via CATALYST are being restored to the DPO. 

 

Production Processing Status 

The rate column will be removed from the processing status tables. 

 

The processing paradigm is changing to process a stream to completion of a month instead of 

processing when data comes in.  Shawn will begin processing the SSF stream today. 

   

System Status   

The DPO restoration effort continues.  151 TB of data have been restored.  Restoring Category 3 

files now.  No projected end date.  Walt will provide an updated list of files to restore to 

Chris by early next week. 

 

There was a pause on the ANGe dropbox earlier today to run the backup that didn’t over a snow 

day. 

 

Nelson and Victor are testing the SGE memory correction.  Perl_Lib 992 will be delivered when 

the testing is complete. 

 

Status of Open SCCRs 

1002 – Delivering an update after the MATCH files are in the new directory in the DPO and 

Tom tests his script. 

999 – Might be updated to include a PR Database modification. 
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937 – Updating the Known Issues and Workarounds document.  

996 – In development.  Comparing testing results against the old database.  New target delivery 

date of March 11. 

997 – New target delivery date of March 11.  

1000 – New target delivery date of March 11. 

990 – CM testing is complete.  Will be released in conjunction with Perl_Lib 992 and Inversion 

973.  

992 – Will be delivered after testing of the SGE memory fix is complete. 

973 – May need to update a script due to a change in Perl_Lib.  

979 – Waiting for ephemeris and attitude files to be restored to the DPO to continue testing. 

919 – Hard to determine the cause of the comparison differences since Rita can’t run on AMI-P.  

Need a follow-up meeting with Jonathan, Sunny, Walt, Tammy, and maybe Rita to discuss 

this delivery. 

1001 – Compiling test package.  On track for delivering on March 14. 

960 –Received one month of VIIRS Collection 1.1 data today.  New target delivery date of 

March 21. 

959 – New target delivery date of April 25.   

940 – On hold for now.  Delivery date TBD. 

974 – No updates.   

975 – No updates. 

        

Action Items  

 

Table 1. Open Action Items 

Item 

Number 

Action Item Date Status 

2014 -01 Definition and implementation of 

Configuration Codes 

02/05/2014 Lisa will work with Pam M. to determine the 

configuration code information to post on the 

Web. 

02/19/2014 Pam M. has provided feedback.  Lisa will 

incorporate Pam’s comments and then provide 

the document to Jonathan, Norman, and Dave 

D. to review. 

03/05/2014 Lisa needs to update the CC document to 

include a delta delivery scenario. 

 

 

 

 

 

Meeting ended at 1:55 p. m. 


